## **Evolutionary Behavioral Sciences**

# Why Human Behavioral Ecology Needs Behavioral Genetics: The Problem of Phenotypic Gambit

Janko Međedović

Online First Publication, January 2, 2023. https://dx.doi.org/10.1037/ebs0000321

#### CITATION

Međedović, J. (2023, January 2). Why Human Behavioral Ecology Needs Behavioral Genetics: The Problem of Phenotypic Gambit. *Evolutionary Behavioral Sciences*. Advance online publication. https://dx.doi.org/10.1037/ebs0000321



© 2023 American Psychological Association ISSN: 2330-2925

### Why Human Behavioral Ecology Needs Behavioral Genetics: The Problem of Phenotypic Gambit

#### Janko Međedović

Institute of Criminological and Sociological Research, Belgrade, Serbia

The phenotypic gambit is one of the crucial assumptions in evolutionary behavioral ecology: it asserts that a phenotypic variation of a trait represents a reasonably adequate estimation of its genetic variation. This is particularly important since one of the main goals of behavioral ecology is to analyze the patterns of natural selection on phenotypic traits—the phenotypic gambit allows only the measurement of phenotypic variance of a trait, without exploring its genetic variance. The phenotypic gambit is crucially dependent on heritability itself-if there are systematic factors that decrease heritability, there is elevated chance for phenotypic gambit to fail. We argue that there is at least one ecological condition, harsh environment, which is related to markedly lower heritability of life history and behavioral traits in humans. Hence, the measurement of phenotypic variance of a trait may generate invalid results, at least for participants who originated from harsh environmental conditions. As a solution we propose the integration of behavioral ecological and behavioral genetic research designs: this integrative design allows researchers to partition the genetic and environmental variances of an examined trait and to obtain genetic correlations between the trait and fitness proxies. Furthermore, it enables that ecological conditions can be analyzed as a moderator in this link. Due to the strengths of integrated design, we believe that this research approach may be highly fruitful for the future studies in evolutionary human sciences.

#### **Public Significance Statement**

Empirical research of behavioral evolution is often conducted only on phenotypes; however, evolutionary processes are unfolding on a genetic level. Observed phenotypic data may not hold on a genetic level, especially in certain ecological conditions. Therefore, we propose closer collaboration between human (evolutionary) behavioral ecology and behavioral genetics and emphasize that this cooperation would benefit and advance both research fields.

*Keywords:* phenotypic gambit, human behavioral ecology, behavioral genetics, fitness, harsh environment, life history, personality

#### Phenotypic Gambit in Human Behavioral Ecology

Human behavioral ecology represents a unique scientific discipline which bridges natural and social sciences. It is an extension of behavioral ecology-a subdiscipline of evolutionary biology aimed to explore morphological, physiological, and behavioral traits in the light of evolutionary theory (Birkhead & Monaghan, 2010). The central term in human behavioral ecology, similarly to evolutionary biology, is fitness. It is defined broadly as a probability of transferring individual genes from one generation to the next (Hunt & Hodgson, 2010); this probability has the highest contingency to the lifetime number of offspring-this is why reproductive success is most frequently used as the main fitness proxy. The basic method of human behavioral ecology is based on analyzing the relations between behavioral traits of interest and fitness-if such relations exist, they suggest

Janko Međedović D https://orcid.org/0000-0001-6022-7934

The author would like to express his gratitude to Denis Bratko for his constructive and helpful comments on the present manuscript.

The author declares no conflict of interest.

Correspondence concerning this article should be addressed to Janko Međedović, Institute of Criminological and Sociological Research, Gračanička 18, 11000 Belgrade, Serbia. Email: janko.medjedovic@fmk.edu.rs

that examined behavioral traits are under natural selection. Hence, the empirical research of current selection regimes on human behavioral, morphological, and physiological traits represents a fundamental research goal in human behavioral ecology-this is the only empirical way to analyze adaptive function of the trait (in an evolutionary sense) and to estimate the effects of selection to the future mean population levels of a trait. The main assumption of behavioral ecology is that populations adapt to their local environments-this means that a certain behavioral trait is not universally adaptive, but it can increase fitness only in certain ecological conditions. Having in mind that humans populate various ecological niches, human behavioral ecology can explain the diversity of human behavior both within and between populations.

Note that the definition of selection is based solely on the covariation between a trait and fitness-if this association exists, we may conclude that natural selection acts upon a trait. However, this does not mean that a trait will respond to selection. In fact, response to selection may occur only if there is genetic covariation between a trait and fitness. However, most of the research in the field of human behavioral ecology is conducted only on the phenotypic measures of analyzed traits. The underlying rationale for this is an assumption of phenotypic gambit (Grafen, 1984; Krebs & Davies, 1978). This assumption states that the phenotypic variance of a trait is reasonably valid estimation of its genetic variation (van Oers & Sinn, 2011); furthermore, it implies that the genetic architecture of a trait does not represent a constraint for its phenotypic evolution (Fawcett et al., 2013). As a result, researchers may explore only phenotypic trait values and their relations with evolutionary fitness (Rubin, 2016). If the covariation between a trait and fitness is detected, then researchers can reasonably assume that this covariation is partly based on a trait's genetic variance; consequently, we can expect phenotypic changes in the mean levels of the trait across generations due to directional selection or reduction in its variation in a case of stabilizing selection. The phenotypic gambit is one of the fundamental assumptions in behavioral ecology-it allows that empirical studies do not estimate genetic variation directly, they can rely solely on phenotypic information instead. Indeed, there are empirical findings which show that phenotypic gambit may be a solid assumption in both humans (Sodini et al., 2018) and other animals (Brommer & Kluen, 2012). On the other hand, the phenotypic gambit has been criticized on various grounds, especially because it oversimplifies the relations between the genotype and phenotype (Dall et al., 2019). Furthermore, the assumption that genetic architecture does not constrain the evolution of a trait may too be invalid (Hadfield et al., 2007; Rubin, 2016).

In the present manuscript, we aim to question the phenotypic gambit on the ground of heritability itself. Heritability is usually defined as a proportion of a phenotypic trait's variation which is explained by its genetic underpinnings. It is a population statistic that is sensitive to cohort characteristics, ecological conditions, and even the age of individuals in which is estimated. Heritability can be estimated using various methods; some of the most frequent ones are twins and family studies (where heritability is estimated via comparing phenotypic correlations between the individuals varying in genetic resemblance; e.g., Posthuma et al., 2003), pedigree data (also known as "animal" model, where heritability is derived from the similarity in phenotypic traits of individuals sharing the same biological lineage; e.g., Bochud, 2012; Milne et al., 2008; Schwartz et al., 2015) or the samples of unrelated individuals where a large number of genetic polymorphisms is used to infer heritability genome-wide association studies or GWAS and genome-wide complex trait analysis or GCTA (e.g., Yang et al., 2011). Although natural selection potentially may act on any trait, a response to selection will crucially depend on trait's heritability because evolutionary changes are based on the change of the frequency of gene alleles which contribute to the trait's phenotypic expressions. However, heritability estimates tend to be variable: this fact represents a problem for phenotypic gambit per se, because in conditions of low heritability, the phenotypic gambit has a higher probability to fail (van Oers & Sinn, 2011). Furthermore, we would like to emphasize an additional condition which may systematically deplete heritability, the one which may be especially important in a behavioral ecological research-harsh environment. First, we would like to illustrate the problem of phenotypic gambit by analyzing heritability estimates in life history and behavioral traits.

#### The Problem of Phenotypic Gambit: Variation in Heritability Coefficients

#### Heritability Coefficients of Life History Traits

Behavioral ecologists believe that in order to understand how individuals maximize fitness we should measure various indicators related to reproductive, parental, or longevity-related behavior. The most frequently measured traits and events are reproductive success, age of first and last reproduction, pubertal timing, onset of sexual behavior, number of sexual partners, and parental investment (Belsky, 2012; Mell et al., 2018; Xu et al., 2018). A conceptual framework which analyzes the relations between these indicators using evolutionary theory is called life history theory (Bolund et al., 2016; Roff, 2002). Life histories of individuals are heavily dependent on the ecological conditions. One of the main ecological determinants of life history dynamics is environmental harshness. Generally, it is assumed that harsh, stressful, low resource, and hostile environments are related to life history patterns aimed at maximize fertility: earlier maturation, onset of sexual behavior and first reproduction, short-term mating, higher reproductive success, followed by a decrease in parental investment and overall health. In contrast, a beneficial and supporting environment, abundant with resources, is associated with the opposite trend: delayed maturation, onset of sexual behavior and first reproduction, long-term mating patterns, lower number of offspring with elevated parental investment, and increased levels of health and longevity. The former pattern of life history indicators is labeled as a "fast" life history pathway, while the latter is labeled as "slow" (Del Giudice et al., 2015). Various characteristics of harsh environment have been found to positively associate with the fast life history trajectory: frequency of violent crimes (Copping & Campbell, 2015), higher mortality rates and lower life expectancy on birth (Griskevicius et al., 2011; Wilson & Daly, 1997), lower socioeconomic status (Sheppard et al., 2016), violent intergroup conflict (Međedović, 2019a), the absence of a father (Webster et al., 2014), lack of maternal sensitivity (Dunkel et al., 2015), troubled family relations (Chisholm et al., 2005), and so on. It is important to note that the application of life history theory in explaining human individual differences and especially the existence of fast–slow continuum has been recently criticized (Međedović, 2020, 2021; Sear, 2020; Stearns & Rodrigues, 2020; Zietsch & Sidari, 2020). Still, other authors believe that the slow–fast dimension of life history dynamics is still useful in interpreting the data, although the scholars should invest more effort to refine life history framework and provide more complex models for its application in humans (Del Giudice, 2020; Galipaud & Kokko, 2020).

Life history traits are heritable, although the proportion of genetic variance in life history traits is lower than the one in morphological or behavioral traits in both humans (Stearns et al., 2010) and other animals (Visscher et al., 2008). Certainly, the most central life history trait is reproductive success itself-the number of biological children an individual has also labeled as fertility. Family and pedigree studies yielded similar results regarding the heritability of reproductive success. First, most of the twin studies found additive genetic variation in reproductive success (e.g., Colodro-Conde et al., 2013; Kohler et al., 2002; Polderman et al., 2015; Silventoinen et al., 2013; reviewed in Mills & Tropf, 2015). Furthermore, they revealed that the heritability of reproduction success highly varies between the cohorts of participants (Briley et al., 2015) or estimates of fertility in different stages of ontogeny (Rodgers et al., 2007)-this variation ranges from zero to moderate effect sizes.

Pedigree studies estimates are lower in magnitude (e.g., Bolund et al., 2013 [fertility is estimated as the number of grandchildren in this study]; Bolund & Lummaa, 2017) but importantly, the range of heritability estimates is also high—when comparing fertility in different life stages (Pettay et al., 2008) or different samples (Milot et al., 2011; Pettay et al., 2005), coefficients range from nonsignificant to moderate ones.

Quite similar empirical data exist for other life history traits. Data also show that heritability coefficients of the age of first reproduction highly vary—from near-to-zero to moderate effect sizes (Mills & Tropf, 2015). Both twin (Briley et al., 2015; Silventoinen et al., 2013; Tropf, Barban, et al., 2015) and pedigree methods converge to this conclusion (Bolund et al., 2013; Bürkli & Postma, 2014). The data on age of menarche are a bit different: estimates show higher heritability, compared to reproductive success and age of first reproduction, which ranges from 50%to 70% depending on the method of estimation and sample characteristics (Sørensen et al., 2013; Sulem et al., 2009; van den Berg & Boomsma, 2007). The heritability of other life history traits has been examined as well; however, the research on other traits is still quite scarce so we cannot reliably estimate the variability of heritability coefficients. Nevertheless, significant heritability coefficients have been found in breastfeeding (as an indicator of parental investment), the onset of sexual behavior, the number of sexual partners, and the probability of being married (Cherkas et al., 2004; Colodro-Conde et al., 2013; Trumbetta et al., 2007; Waldron et al., 2007).

#### Explaining Heritability Variation in Life History Traits: Implications for Current Life History Evolution

The variation in heritability coefficients certainly has multiple causes; some of the frequently detected parameters which influence this variation are the age and sex of participants (Mills & Tropf, 2015). There are environmental causes in this variation as well. It has been noted that heritability of fertility and age of first reproduction changes across the cohorts in the late 19th and whole 20th century-generally the data point to the conclusion that heritability is higher when demographic transitions started to unravel (Kohler et al., 2002); for example, heritability is higher in individuals born between 1936 and 1955 compared to the 1920-1935 cohort (Briley et al., 2015; however, in the 1956–1970 cohort the heritability of fertility persisted but not the one of the age of first reproduction). Heritability of fertility was also higher in urban, compared to rural environments (Bras et al., 2013). One of the explanations for these results is that in demographic transition populations (characterized by lower fertility partially caused via accessible methods of birth control), urban environments, and later 20th-century generations there was enhanced choice in making reproductive decisions (Briley et al., 2015; Udry, 1996). Perhaps it sounds counterintuitive that the elevated choice in fertility-related behavior gives rise to the genetic underpinnings of the same behavior. However, this explanation becomes apparent if we note the following: (a) the role of social and environmental factors in a broader sense has decreased in these contexts, which allowed genetic mechanisms to influence behavior in a greater extent; (b) motives for childbearing, including the number of desired children are heritable psychological traits as well (Miller et al., 2010; Pasta & Miller, 2000), hence, when individuals have a choice to make their reproductive decisions, this choice is partly a result of underlying genetic variation. The data of shared genetic variation between fertility motivation and observed number of children are clearly in line with the previous explanation (Rodgers et al., 2001).

The data on elevated heritability of fertility in urban environments can be explained by other processes postulated in behavioral genetics that have high relevance for behavioral ecology as well: genotype-environment correlations or rGEs (Meek et al., 2013). There are three types of these correlations: passive, evocative, and active (Plomin et al., 1977). Passive correlation describes the congruence between parental genotype and parental behavior toward offspring; evocative correlation highlights the responses from children toward their environment that further guides their development. Finally, active correlation describes the tendency of individuals to search for new environments that suits their genetic potentials; therefore, changing the environments may lead to a better match between ecological conditions and genetic dispositions. Furthermore, if we choose environments in order to be suited for our genetic dispositions, this would enable higher genetic expressions on behavior, and consequently the increase in heritability estimates. This is why active genotypeenvironment correlation can explain higher heritability of fertility in urban environments, especially in conditions of elevated migration from rural to urban ecologies. The conceptual benefits of active genotype-environment correlation for behavioral ecology are apparent-if changing environments leads to better functioning in new ecological conditions, this behavior may be viewed as an adaptation. However, as far as we are aware, there is no empirical research in humans that examine active genotype-environment correlation in the context of biological adaptation so far-this remains a fruitful topic for future research.

The magnitude of heritability coefficients is linked to environmental harshness as well broadly viewed as the level of deprivation, lack of resources, hostile, and stressful environments versus beneficial, resource-abundant, and supporting environments. It has been noted that the genetic variation in number of children and grandchildren is higher in landowning compared to landless families (Bolund & Lummaa, 2017). Intriguing findings are obtained on the heritability of first sexual intercourse-genetic variation explained 39% of the onset of sexual behavior's phenotypic variation in females without the history of child sexual abuse; however, in women who reported sexual abuse in childhood, the genetic influences on the age of first intercourse drops to zero (Waldron et al., 2008). Hence, the percentage of genetic variation in phenotypic values of life history traits is higher in favorable environments, a finding which is obtained in other animals as well (Visscher et al., 2008). Note that the levels of harshness are probably related to the level of individual freedom discussed earlier: more supporting and beneficial environments are likely the same ones which provide higher freedom to make choices about sexual and reproductive behavior.

Thus, the variation in heritability of life history traits may represent one of the constraints for their evolution, and it may cause the failure of phenotypic gambit. There are empirical data which suggest that this indeed may be the case. For example, one study found low but significant positive association between breastfeeding and reproductive success (interestingly, life history theory would predict a negative association between these traits<sup>1</sup>)—this finding alone could suggest that breastfeeding may be under positive directional selection; nevertheless, the genetic association between these two traits was not statistically significant (Colodro-Conde et al., 2013). A more striking example comes from the study which investigated sex differences regarding selection on several life history traits (Bolund et al., 2013). Phenotypic analysis showed that sex moderated relations between age of first and last reproduction, and fitness (i.e., reproductive success)-this finding implies sexual selection on these traits which may result in heightened differences in timing of reproduction between males and females. However, genetic correlations between life history traits and fitness were quite similar in both sexes, the finding that suggests an opposite process-a convergent life history evolution in males and females. Another study (Beauchamp, 2016) showed positive associations between body mass and fitness and negative between female height and fitness, but there were no genetic associations between these traits; phenotypic gambit held only for educational level which showed negative genetic and phenotypic associations with fitness. Finally, there is even evidence of a larger discrepancy between genotypic and phenotypic levels of analysis-associations which have opposite signs. Sanjak et al. (2018) obtained some converging links between life history traits and fitness (e.g., negative association between age of first birth and fitness) but they found positive relation between age of menopause and fitness on the phenotypic and negative on the genetic level.

Hence, there are empirical data that cast a doubt on the phenotypic gambit's reliability. Furthermore, there is reason to doubt phenotypic gambit even when there are mirroring associations on phenotypic and genetic levels. One of the key assumptions of life history theory is related to local ecologies: a harsher and depriving environment should facilitate the evolution of phenotypes which contribute to reproductive output and early fertility, including morphological, physiological, behavioral, and life history traits in a narrow sense. However, previously described data on life history traits' heritability showed lower heritability coefficients in the conditions of a harsh environment. Therefore, it is questionable whether genetic correlations between life history traits and fitness would hold in subpopulations characterized by harsh environmental conditions-the mere lack of genetic variation may constrict covariation with fitness as well. In these cases, a fast life history phenotypic response in a harsh environment would not be a consequence of genetic evolution

<sup>&</sup>lt;sup>1</sup> Breastfeeding is a form of parental investment; one of the major evolutionary tradeoffs is the one between quality and quantity of offspring—this tradeoff implies that parents cannot invest as much if they have higher number of offspring compared to fewer offspring. Parental investment can evolve if individuals with lower number of offspring can have off-spring with elevated quality by increased investment in them. In the life history context, fast life history trajectory would be depicted by higher reproductive success and lower parental investment; inverse pattern should characterize slow life history. The cited study shows that this prediction cannot hold in populations where fertility is generally low —quantity-quality tradeoff can be more easily found in natural fertility populations (Gillespie et al., 2008; Meij et al., 2009).

but the processes which act on an ontogenetic level like developmental and behavioral plasticity (Nettle & Bateson, 2015).

#### The Case of Behavioral Traits—Personality and Cognition

A personality trait is any behavioral pattern that is relatively stable in time, and that shows considerable inter-individual differences in a population. There are many taxonomies of human personality traits; one of the most prominent is the Big Five personality model (John et al., 2008) which consists of five traits: Extraversion (sociability, activity, positive affectivity), Neuroticism (negative emotionality), Agreeableness (cooperation, empathy, altruism), Conscientiousness (orderliness, behavioral control), and Openness to experience (inquisitiveness, creativity). Personality traits are adaptive behavioral dispositions (at least at certain levels of traits' expressions)-a fact which is reflected in their relations with fitness in both humans (Allen & Robson, 2018; Gurven et al., 2014; Međedović et al., 2018) and other animals (Smith & Blumstein, 2008). In fact, it is shown that every one of the major human personality characteristics is associated with fitness, although the obtained links may vary as a function of sample characteristics (Penke & Jokela, 2016). Personality traits show moderate heritabilityaround 40% (Vukasović & Bratko, 2015); this fact, combined with their association with fitness, marks their potential to evolve under natural selection.

Unfortunately, there are only a few studies which examine if the link between personality and fitness exists on genetic level as well. In fact, some data showed that phenotypic gambit may apply to human personality: certain findings showed the same phenotypic and genetic associations between personality traits and fitness (Berg et al., 2016; Briley et al., 2017; the matrix of phenotypic associations between personality traits and fitness in the latter study was obtained via personal communication with the first author). However, it is highly debatable if these associations would hold in every sample. The heritability of personality is highly variable and depends on numerous characteristics of the analyzed samples (Bratko et al., 2017), including the age of the participants (Kandler et al., 2021). Furthermore, environmental harshness turned out to be an important determinant of heritability in personality as well: it has been found that the heritability coefficients are lower in participants who received more negative and stressful parental behavior (Krueger et al., 2008). There are findings showing that the heritability of the Emotional Dysregulation trait (which is similar to the trait Neuroticism) drops almost to zero in environments with high family conflict (Jang et al., 2005). The existing data imply that the heritable variation in personality may be reduced in conditions of environmental harshness.

The hypothesized role of the environment in the heritability of behavioral traits is even more convincing for cognitive abilities. Intelligence is a broad term to capture cognitive abilities related to problem solving and it has a profound influence on various behaviors and life outcomes. Furthermore, intelligence is a trait which probably has the highest heritability among the human behavioral characteristicsestimates even go to 80% (Sauce & Matzel, 2018). While it is quite plausible to hypothesize that cognitive abilities evolved as an adaptation, there is an ongoing debate whether intelligence is adaptive in contemporary humans, due to a mismatch between adaptations and environment. This debate is reflected in empirical data as well: both negative (Reeve et al., 2013; Shatz, 2008) (Kolk and positive & Barclay, 2021: Međedović, 2017) associations were detected between the phenotypic scores of intellectual performance and fitness. Hence, examining the current evolution of intelligence remains an intriguing and exciting task. Various factors contribute to the variation of heritability estimates in intelligence; for example, age seems to be an important cause of heritability coefficients' increase in cognitive abilities (Haworth et al., 2010). Furthermore, there are differences in heritability of intelligence in different environmental conditions-the data show that heritability estimates are lower in individuals who grew up in the conditions of economical poverty (Tucker-Drob et al., 2013). There are studies estimating the heritability of intelligence to be 72% in wealthy, compared to 10% in children living in economic deprivation (Turkheimer et al, 2003). Although, it must be noted that the effect of harsh environment on the heritability estimates of intelligence is more common for some countries than others; for example, the effect is ubiquitous in the United States but practically absent in Australia (Bates et al., 2016; Tucker-Drob & Bates, 2016). Hence, harsh environment may not be necessarily linked to heritability depletion in cognitive abilities and this association may be dependent on other ecological conditions. There are additional critiques of this assumption as well—some data suggest that lower heritability estimates in poorer environments are not the consequence of lower additive genetic variation but elevated shared environmental variance (Hanscombe et al., 2012). Nevertheless, such a high range in intelligence's heritability can represent a major obstacle in the research on its contemporary evolution if only phenotypic scores of cognitive abilities are analyzed: low heritability decreases the chances that the covariation between the trait and fitness will be found on a genetic level. Consequently, it diminishes the possibility that the trait would respond to natural selection.

## Particularly Vulnerable Traits—Adaptations to Harsh Environment

Behavioral ecologists often explicitly analyze ecological conditions as factors which moderate the links between a trait and fitness, that is, there are traits which may be adaptive only in certain conditions. In fact, there are behavioral traits which are assumed to be adaptive especially in harsh environments. An example of such a trait is psychopathy-it represents a behavioral syndrome consisting of several traits: diminished tendency to generate negative emotions (especially fear, guilt, and empathy), manipulative and deceitful behavior, and lower behavioral control based on impulsivity and disinhibition (Hare, 2003; Lilienfeld & Andrews, 1996; Patrick et al., 2009). Similarly to other personality traits, psychopathy was shown to have moderate heritability (Tuvblad et al., 2019). Evolutionary ecological analysis suggested that psychopathy may positively contribute to evolutionary fitness although the adaptiveness of psychopathy may be dependent on its frequency in a population—psychopathic phenotypes may be adaptive only if they are rare (Mealey, 1995). Furthermore, it is assumed that psychopathy can increase fitness especially in the harsh environmental conditions (Glenn et al., 2011). Psychopathic affectivity may represent a trait which enables individuals to overcome

adaptive challenges more successfully in such environment because it elevates stress tolerance. Despite the fact that psychopathy is mainly viewed as a maladaptive, even pathological trait, empirical data confirmed behavioral ecological predictions: manipulative and affective traits were found to elevate reproductive fitness (Međedović, 2019b; Međedović & Petrović, 2019) and adaptive potentials of psychopathy are indeed more expressed in a harsh, depriving, and stressful environment (Mededović, 2019c; Međedović et al., 2017). However, the abovementioned associations were obtained only on the phenotypic level. Since the possibility that psychopathic traits have lower heritability in a harsh environment is at least plausible, we can pose a question whether these links can be replicated on a genetic level as well. Unfortunately, there are no empirical research studies that examine genetic associations between psychopathy and fitness, especially the ones which explore these links in various environments so far.

#### A Possible Solution: Integration of Behavioral Ecology and Behavioral Genetics

#### Behavioral Genetics and Its Application to Human Behavioral Ecology

It seems that the phenotypic gambit cannot be sustained in all circumstances; furthermore, there are environmental conditions, such as harsh environment, which elevate the probability of phenotypic gambit's failure. The solution seems apparent: behavioral ecologists need genetic information which could allow them to more validly and reliably infer if the trait under natural selection would indeed respond to selection. Molecular genetics has been largely unsuccessful in determining gene variants which participate in a phenotypic expression of behavioral traits (Jarnecke & South, 2017), mostly due to their highly polygenic nature; gene-candidate studies showed to be particularly prone to fail in replicating the findings about the specific genetic polymorphisms related to behavioral traits, while GWAS studies have higher chance to find and replicate the findings on molecular genetic basis of behavior (Chabris et al., 2015). Among several methods, behavioral genetics, as a quantitative genetics applied to behavior, seems to have the highest analytic potential since it can partition the phenotypic variation of a trait into the genetic and environmental variance components, allowing analyses to be performed on both variances (Plomin et al., 2008). Furthermore, compared to some other methods of heritability estimation, behavioral genetics provides higher genetic variation estimates: other methods such as GWAS are characterized by missing heritability, which results in lower heritability estimates (Manolio et al., 2009; although it must be said that GWAS studies have substantial contribution to understanding genetics of reproductive-related traits: Gajbhiye et al., 2018). The integration of behavioral genetics in evolutionary human sciences has been proposed on several previous occasions (see Barbaro et al., 2017). These appeals came mostly from evolutionary psychology (Zietsch et al., 2015); however, due to their theoretical assumptions (e.g., evolutionary mismatch hypothesis; Li et al., 2018), evolutionary psychologists do not aim to empirically estimate current selection regimes on behavioral traits (i.e., they do not measure current fitness). Hence, we are focused on ways behavioral genetics can help behavioral ecologists analyze the relations between behavioral traits and fitness.

Behavioral genetics uses information about the genetic similarity between individuals to estimate heritability. The most prominent behavioral genetic methodology is based on the examination of the phenotypic traits in twin pairs: heritability is estimated by comparing correlations of phenotypic traits in monozygotic and dizygotic twins (Plomin et al., 2008). But more importantly, this method allows for decomposing the phenotypic trait's variation into genetic, shared environmental (the environment which is common for both twins such as family context), and unique environmental variation (the events and experience which is unique for every member of a twin pair). Shared environment results in heightened similarity between the twins, whereas non-shared environment contributes to the differences between the twin-pair members. Hence, every analysis may be performed on phenotypic, genetic, and environmental variation of examined phenotypical measures (shared environmental variance is quite low for many traits, therefore, for such traits only unique environmental variation is suitable for separate analyses).

The main benefit of behavioral genetics is the ability to perform direct testing of phenotypic gambit. The majority of behavioral genetics

studies compare phenotypic and genetic correlations of the examined variables. But in behavioral ecology, the baseline research design should be a bit more complex. The integration of behavioral ecology and behavioral genetics must include at least four sets of variables: (a) the traits which represent target concepts of the researchlife history, physiological, morphological, or behavioral traits; (b) fitness proxies-reproductive success (number of children and grandchildren), longevity, parental investment, and so on; (c) genetic information-in this case, represented as genetic variation in general, not specific alleles or other polymorphisms; (d) environmental conditions-harsh environment or other ecological variation of interest. Thus, the method is not simply based on conducting separate analyses on genetic and environmental variation of certain traits-the crucial hypotheses involve testing the environmental conditions as a moderator in the link between certain traits and fitness. This means that the analysis must involve two steps: first, a researcher removes the environmental variation of analyzed variables, leaving only genetic variation for further examination; afterwards a researcher should analyze if the genetic links between the trait and fitness exist in different environments (e.g., harsh and beneficial). For example, if we want to obtain unambiguous evidence of natural selection's influences on psychopathy we analyze the genetic link between psychopathy and reproductive success; subsequently, we examine if this (genetic) link holds for various environmental conditions or if it exists only in certain ecologies.

Basic conceptual frameworks for these analyses already exist in behavior genetics. The hypothesis that genetic links between a trait and fitness exist only in certain environments is congruent with the concept of gene-environment interaction (Manuck & McCaffery, 2014; Shanahan & Hofer, 2005). It is often defined as genetic control of sensitivity to various environments, or equivalently, as the environmental control of differential gene effects. In behavior genetics research, the particularly important design incorporates genetic variation operationalized as a latent variable (e.g., additive genetic variance) with environment as a measured variable (e.g., socioeconomic status, childhood stress, social ostracism). Behavior geneticists typically examine if heritability coefficients for a given trait vary across different environments (Purcell, 2002). In a behavioral ecological context, we could examine genetic associations between traits and fitness across different environments.

This said, it is important to mention the limitations of behavioral genetics as well. The most common criticisms point to two problems; firstly, the assumption of shared environment may be invalid - do parents have the same behavior dizygotic toward twins, especially ones? Because if they not, the shared environmental influences are not the same for dizygotic twines as the model expects. Secondly, twin samples may not be representative for the whole population; hence a problem of findings' generalizability emerges (e.g., Burt & Simons, 2014; Jarnecke & South, 2017). Another problem refers to assortative mating: if couples' pairing is based on similarity in certain traits, this may influence biometric parameters by increasing the heritability estimates (Ruby et al., 2018). However, other researchers pointed out that possible violations of biometric modeling assumptions do not invalidate the results of behavioral genetic analyses. For example, it is shown that the hypothesis of equal shared environment for monozygotic and dizygotic twins is rarely violated, and even if it is it has very low and unsystematic effect on heritability estimates (for a detailed discussion on this topic, see Barnes et al., 2014). Still, some specific aspects of generalizability criticism can be especially important for behavioral ecology: by examining only twins, behavioral genetic studies exclude families with low fertility; furthermore, dizygotic twinning is genetically based which means that including dizygotic twins in a study sample may produce biased estimations of fertility-related traits (Tropf, Stulp, et al., 2015).

#### Additional Potentials of a Behavioral Ecology/ Genetics Integrated Design

The partition of a trait's variation into genetic and environmental ones is the most basic behavioral genetic design. The methodology provides more complex designs based on structural modeling that are of clear importance for behavioral ecology. Therefore, although we were focused on a harsh environment as a factor which decreases heritability and consequently questions phenotypic gambit, we would like to provide other reasons for the integration of behavioral ecology and genetics. We do not aim to present a comprehensive review, but examples of heuristic powers embedded in integrated design.

In behavioral ecology it is very rare to explore the link between a single trait and fitness-multivariate designs are frequently applied, the ones where associations between the suits of traits and fitness are estimated simultaneously (Sinervo & Calsbeek, 2010). This is of utmost importance since various characteristics of individual differences are related between themselves, both on a phenotypic and genetic level. The latter associations are the result of genetic pleiotropy—a situation where a single genetic variant or a suit of variants are associated with several phenotypic traits. Pleiotropy has an important implication for behavioral ecological analysis; for example, a trait does not need to be directly genetically related to fitness in order to be under natural selection-it may be affected by selection if it is genetically associated with other traits linked with fitness. An example comes from one of the rare research which explored genetic associations between personality traits and fitness: the data showed that the Neuroticism trait is not related to reproductive success on a genetic level; however, Neuroticism was genetically negatively associated with Extraversion, a trait found to elevate the number of children on a genetic level (Berg et al., 2016). Therefore, if we analyzed only bivariate genetic associations between personality traits and fitness we may erroneously conclude that Neuroticism is selectively neutral; however, it may be under negative directional selection via its genetic associations with Extraversion. Multivariate models provide another advantage besides biologically more valid estimation of traits' evolution due to the presence of pleiotropy and correlational selection-exploration of multivariate genetic constraints on traits' change under natural selection (Walsh & Blows, 2009).

The problem of causality is present in behavioral ecological research as well—the most valid information on selective forces acting on a trait should demonstrate that trait affects fitness, not vice versa (but vice versa is plausible as well, for example, having children may change personality traits of parents: Jokela et al., 2009). Hence, longitudinal designs enable more convincing data in behavioral ecology, similar to other scientific disciplines. Behavioral genetics provide an additional piece of information to these designs: by using Cholesky models behavioral geneticists may analyze if genetic variation of a trait obtained at an earlier age can predict variation in fitness at a later age (preferably in a postreproductive period of ontogeny). For example, Rodgers et al. (2007) found that genetic factors which are positively associated with early fertility (at the age of 20 years) are negatively associated with later fertility (at the age of 30 and 35 years). This finding suggests that the same genetic variants may have opposite effects on fitness in different stages of ontogeny.

There is another exciting paradigm in animal behavioral ecology which examines social evolution of the traits—it is labeled as indirect genetic effects or interacting phenotypes (Bailey et al., 2018; Wolf & Moore, 2010). Indirect genetic effects emerge if the phenotype of a focal individual is influenced by the genotype of its social partners. Hence, the environmental variation examined in this framework is social, not the physical environment-in principle, every individual that has interactions with the focal individual can be analyzed as a source of indirect genetic effects. One of the apparent social partners analyzed in empirical research are mating partners (Danielson-François et al., 2009) and parental dyads (Head et al., 2012). Unfortunately, interacting phenotypes are not yet studied in human behavioral ecology, at least as far as we are aware. This is unfortunate because they provide a unique theoretical perspective which may provide more valid and precise estimations of traits' evolution since interactive phenotypes can bias evolutionary trajectories and elevate traits' response to selection (Bailey et al., 2018). Behavioral genetics can provide an adequate platform for examining indirect genetic effects in humans. The logic of a research design is similar to previously described studies focused on environmental harshness. We could use twin designs to decompose the variation of a trait into genetic and environmental components; afterwards we can analyze if the genetic variation of a trait is related to fitness of their mating partners-for example, their fertility, longevity, or parental investment.

#### **Concluding Remarks**

Behavioral genetics and behavioral ecology had distinct developmental trajectories as scientific disciplines and even today they remain relatively separate. This state of affairs is not surprising since the main research focus and questions are different in these two disciplines. In the present manuscript, we focused on a problem of phenotypic gambit as a potentially critical condition which may jeopardize the findings in behavioral ecology; however, the integration of two fields has an extraordinary explanatory potential in general and it can help both fields to further develop and grow. Behavioral genetic methodology can help human evolutionary ecology to obtain unambiguous answers of selective regimes acting on phenotypic traits, and to pose more complex questions regarding the interactions between individuals and their ecologies, physical and social environments. On the other hand, behavioral ecological approach can open new doors of scientific inquiry for behavioral geneticists-the latter mostly estimated genetic and environmental variation of phenotypic traits and analyzed its consequences so far. Behavioral ecology can help them to study the causes of this variation and its evolutionary dynamics; this can provide deeper insight into the complex evolution of behavioral traits and help estimate future changes in phenotypic characteristics. Therefore, evolutionary social sciences would gain large benefits if these two research fields would integrate: this may help in opening new and even more exciting chapters in the story of contemporary human evolution.

#### References

- Allen, M. S., & Robson, D. A. (2018). A 10-year prospective study of personality and reproductive success: Testing the mediating role of healthy living. *Psychology & Health*, 33(11), 1379–1395. https:// doi.org/10.1080/08870446.2018.1498499
- Bailey, N. W., Marie-Orleach, L., & Moore, A. J. (2018). Indirect genetic effects in behavioral ecology: Does behavior play a special role in evolution? *Behavioral Ecology*, 29(1), 1–11. https://doi.org/10 .1093/beheco/arx127
- Barbaro, N., Boutwell, B. B., Barnes, J. C., & Shackelford, T. K. (2017). Genetic confounding of the relationship between father absence and age at menarche. *Evolution and Human Behavior*, 38(3), 357– 365. https://doi.org/10.1016/j.evolhumbehav.2016 .11.007
- Barnes, J. C., Wright, J. P., Boutwell, B. B., Schwartz, J. A., Connolly, E. J., Nedelec, J. L., & Beaver, K. M. (2014). Demonstrating the validity of twin

research in criminology. *Criminology*, *52*(4), 588–626. https://doi.org/10.1111/1745-9125.12049

- Bates, T. C., Hansell, N. K., Martin, N. G., & Wright, M. J. (2016). When does socioeconomic status (SES) moderate the heritability of IQ? No evidence for g × SES interaction for IQ in a representative sample of 1176 Australian adolescent twin pairs. *Intelligence*, 56, 10–15. https://doi.org/10.1016/j .intell.2016.02.003
- Beauchamp, J. P. (2016). Genetic evidence for natural selection in humans in the contemporary United States. *Proceedings of the National Academy of Sciences of the United States of America*, *113*(28), 7774–7779. https://doi.org/10.1073/pnas .1600398113
- Belsky, J. (2012). The development of human reproductive strategies: Progress and prospects. *Current Directions in Psychological Science*, 21(5), 310– 316. https://doi.org/10.1177/0963721412453588
- Berg, V., Lummaa, V., Rickard, I. J., Silventoinen, K., Kaprio, J., & Jokela, M. (2016). Genetic associations between personality traits and lifetime reproductive success in humans. *Behavior Genetics*, 46(6), 742– 753. https://doi.org/10.1007/s10519-016-9803-5
- Birkhead, T. R. & Monaghan, P. (2010). Ingenious ideas: The history of behavioral ecology. In D. F. Westneat & C. W. Fox (Eds.), *Evolutionary behavio*ral ecology (pp. 3–15). Oxford University Press.
- Bochud, M. (2012) Estimating heritability from nuclear family and pedigree data. In R. Elston, J. Satagopan, & S. Sun (Eds.), *Statistical human genetics: Methods in molecular biology (methods and protocols)* (Vol. 850, pp. 171–186). Springer/Humana Press. https:// doi.org/10.1007/978-1-61779-555-8\_10
- Bolund, E., Bouwhuis, S., Pettay, J. E., & Lummaa, V. (2013). Divergent selection on, but no genetic conflict over, female and male timing and rate of reproduction in a human population. *Proceedings* of the Royal Society B: Biological Sciences, 280(1772), Article 20132002. https://doi.org/10 .1098/rspb.2013.2002
- Bolund, E., Hayward, A., & Lummaa, V. (2016). Life history evolution, human. In R. M. Kliman (Ed.), *Encyclopedia of evolutionary biology* (pp. 328– 334). Academic Press.
- Bolund, E., & Lummaa, V. (2017). The effects of resource availability and the demographic transition on the genetic correlation between number of children and grandchildren in humans. *Heredity*, *118*(2), 186–192. https://doi.org/10.1038/hdy.2016 .81
- Bras, H., van Bavel, J., & Mandemakers, K. (2013). Unraveling the intergenerational transmission of fertility: Genetic and shared-environment effects during the demographic transition in the Netherlands, 1810–1910. *The History of the Family*, *18*(2), 116–134. https://doi.org/10.1080/1081602X .2013.803491

- Bratko, D., Butković, A., & Vukasović Hlupić, T. (2017). Heritability of personality. *Psihologijske Teme*, 26(1), 1–24. https://doi.org/10.31820/pt.26 .1.1
- Briley, D. A., Harden, K. P., & Tucker-Drob, E. M. (2015). Genotype × cohort interaction on completed fertility and age at first birth. *Behavior Genetics*, 45(1), 71–83. https://doi.org/10.1007/s10519-014-9693-3
- Briley, D. A., Tropf, F. C., & Mills, M. C. (2017). What explains the heritability of completed fertility? Evidence from two large twin studies. *Behavior Genetics*, 47(1), 36–51. https://doi.org/10.1007/ s10519-016-9805-3
- Brommer, J. E., & Kluen, E. (2012). Exploring the genetics of nestling personality traits in a wild passerine bird: Testing the phenotypic gambit. *Ecology* and Evolution, 2(12), 3032–3044. https://doi.org/10 .1002/ece3.412
- Bürkli, A., & Postma, E. (2014). Genetic constraints underlying human reproductive timing in a premodern Swiss village. *Evolution*, 68(2), 526–537. https://doi.org/10.1111/evo.12287
- Burt, C. H., & Simons, R. L. (2014). Pulling back the curtain on heritability studies: Biosocial criminology in the postgenomic era. *Criminology*, 52(2), 223– 262. https://doi.org/10.1111/1745-9125.12036
- Chabris, C. F., Lee, J. J., Cesarini, D., Benjamin, D. J., & Laibson, D. I. (2015). The fourth law of behavior genetics. *Current Directions in Psychological Science*, 24(4), 304–312. https://doi.org/10.1177/ 0963721415580430
- Cherkas, L. F., Oelsner, E. C., Mak, Y. T., Valdes, A., & Spector, T. D. (2004). Genetic influences on female infidelity and number of sexual partners in humans: A linkage and association study of the role of the vasopressin receptor gene (AVPR1A). *Twin Research and Human Genetics*, 7(6), 649– 658. https://doi.org/10.1375/twin.7.6.649
- Chisholm, J. S., Quinlivan, J. A., Petersen, R. W., & Coall, D. A. (2005). Early stress predicts age at menarche and first birth, adult attachment, and expected lifespan. *Human Nature*, 16(3), 233–265. https:// doi.org/10.1007/s12110-005-1009-0
- Colodro-Conde, L., Rijsdijk, F., & Ordoñana, J. R. (2013). The genetic and environmental structure of reproduction-related variables: The case of fertility and breastfeeding. *Twin Research and Human Genetics*, 16(6), 1096–1102. https://doi.org/10 .1017/thg.2013.64
- Copping, L. T., & Campbell, A. (2015). The environment and life history strategies: Neighborhood and individual-level models. *Evolution and Human Behavior*, 36(3), 182–190. https://doi.org/10.1016/ j.evolhumbehav.2014.10.005
- Dall, S. R., McNamara, J. M., & Wilson, A. J. (2019).
  Ultimate (re) thinking for behavioural biology. In D.
  J. Hosken, J. Hunt, & N. Wedell (Eds.), *Genes and*

*behaviour: Beyond nature-nurture* (pp. 11–25). Wiley.

- Danielson-François, A. M., Zhou, Y., & Greenfield, M. D. (2009). Indirect genetic effects and the lek paradox: Inter-genotypic competition may strengthen genotype × environment interactions and conserve genetic variance. *Genetica*, 136(1), 27–36. https://doi.org/10.1007/s10709-008-9297-z
- Del Giudice, M. (2020). Rethinking the fast-slow continuum of individual differences. *Evolution and Human Behavior*, 41(6), 536–549. https://doi.org/ 10.1016/j.evolhumbehav.2020.05.004
- Del Giudice, M., Gangestad, S. W., & Kaplan, H. S. (2015). Life history theory and evolutionary psychology. In D. M. Buss (Ed.), *The handbook of evolutionary psychology. Foundations* (Vol. 1, 2nd ed., pp. 88–114). Wiley. https://doi.org/10.1002/ 9781119125563.evpsych102
- Dunkel, C. S., Mathes, E. W., Kesselring, S. N., Decker, M. L., & Kelts, D. J. (2015). Parenting influence on the development of life history strategy. *Evolution* and Human Behavior, 36(5), 374–378. https:// doi.org/10.1016/j.evolhumbehav.2015.02.006
- Fawcett, T. W., Hamblin, S., & Giraldeau, L. A. (2013). Exposing the behavioral gambit: The evolution of learning and decision rules. *Behavioral Ecol*ogy, 24(1), 2–11. https://doi.org/10.1093/beheco/ ars085
- Gajbhiye, R., Fung, J. N., & Montgomery, G. W. (2018). Complex genetics of female fertility. *npj Genomic Medicine*, 3(1), Article 29. https:// doi.org/10.1038/s41525-018-0068-1
- Galipaud, M., & Kokko, H. (2020). Adaptation and plasticity in life-history theory: How to derive predictions. *Evolution and Human Behavior*, 41(6), 493–501. https://doi.org/10.1016/j.evolhumbehav .2020.06.007
- Gillespie, D. O., Russell, A. F., & Lummaa, V. (2008). When fecundity does not equal fitness: Evidence of an offspring quantity versus quality trade-off in preindustrial humans. *Proceedings of the Royal Society* of London B: Biological Sciences, 275(1635), 713– 722. https://doi.org/10.1098/rspb.2007.1000
- Glenn, A. L., Kurzban, R., & Raine, A. (2011). Evolutionary theory and psychopathy. Aggression and Violent Behavior, 16(5), 371–380. https://doi.org/ 10.1016/j.avb.2011.03.009
- Grafen, A. (1984). Natural selection, kin selection and group selection. In J. R. Krebs & N. B. Davies (Eds.), *Behavioural ecology: An evolutionary approach* (pp. 62–84). Blackwell Scientific Publications.
- Griskevicius, V., Delton, A. W., Robertson, T. E., & Tybur, J. M. (2011). Environmental contingency in life history strategies: The influence of mortality and socioeconomic status on reproductive timing. *Journal of Personality and Social Psychology*, *100*(2), 241–254. https://doi.org/10.1037/a0021082

- Gurven, M., von Rueden, C., Stieglitz, J., Kaplan, H., & Rodriguez, D. E. (2014). The evolutionary fitness of personality traits in a small-scale subsistence society. *Evolution and Human Behavior*, 35(1), 17–25. https://doi.org/10.1016/j.evolhumbehav .2013.09.002
- Hadfield, J. D., Nutall, A., Osorio, D., & Owens, I. P. F. (2007). Testing the phenotypic gambit: Phenotypic, genetic and environmental correlations of colour. *Journal of Evolutionary Biology*, 20(2), 549–557. https://doi.org/10.1111/j.1420-9101 .2006.01262.x
- Hanscombe, K. B., Trzaskowski, M., Haworth, C. M., Davis, O. S., Dale, P. S., & Plomin, R. (2012). Socioeconomic status (SES) and children's Intelligence (IQ): In a UK-representative sample SES moderates the environmental, not genetic, effect on IQ. *PLoS ONE*, 7(2), Article e30320. https://doi.org/10 .1371/journal.pone.0030320
- Hare, R. D. (2003). The Hare Psychopathy Checklist— Revised (2nd ed.). Multi-Health Systems.
- Haworth, C. M., Wright, M. J., Luciano, M., Martin, N. G., de Geus, E. J. C., van Beijsterveldt, C. E. M., Bartels, M., Posthuma, D., Boomsma, D. I., Davis, O. S., Kovas, Y., Corley, R. P., Defries, J. C., Hewitt, J. K., Olson, R. K., Rhea, S. A., Wadsworth, S. J., Iacono, W. G., McGue, M., ... Plomin, R. (2010). The heritability of general cognitive ability increases linearly from childhood to young adulthood. *Molecular Psychiatry*, *15*(11), 1112–1120. https://doi.org/10.1038/ mp.2009.55
- Head, M. L., Berry, L. K., Royle, N. J., & Moore, A. J. (2012). Paternal care: Direct and indirect genetic effects of fathers on offspring performance. *Evolution: International Journal of Organic Evolution*, *66*(11), 3570–3581. https://doi.org/10.1111/j .1558-5646.2012.01699.x
- Hunt, J., & Hodgson, D. J. (2010). What is fitness and how do we measure it? In D. F. Westneat & C. W. Fox (Eds.), *Evolutionary behavioural ecology* (pp. 46–71). Oxford University Press.
- Jang, K. L., Dick, D. M., Wolf, H., Livesley, W. J., & Paris, J. (2005). Psychosocial adversity and emotional instability: An application of gene–environment interaction models. *European Journal of Personality*, 19(4), 359–372. https://doi.org/10 .1002/per.561
- Jarnecke, A. M., & South, S. C. (2017). Behavior and molecular genetics of the Five Factor Model. In T. A. Widiger (Ed.), *The Oxford handbook of the five factor model* (pp. 301–318). Oxford University Press.
- John, O. P., Naumann, L. P., & Soto, C. J. (2008). Paradigm shift to the integrative Big Five trait taxonomy: History, measurement, and conceptual issues. In O. P. John, R. W. Robins, & L. A. Pervin (Eds.), *Handbook of personality: Theory and research* (pp. 114–158). Guilford Press.

- Jokela, M., Kivimäki, M., Elovainio, M., & Keltikangas-Järvinen, L. (2009). Personality and having children: A two-way relationship. *Journal* of Personality and Social Psychology, 96(1), 218– 230. https://doi.org/10.1037/a0014058
- Kandler, C., Bratko, D., Butković, A., Hlupić, T. V., Tybur, J. M., Wesseldijk, L. W., de Vries, R. E., Jern, P., & Lewis, G. J. (2021). How genetic and environmental variance in personality traits shift across the life span: Evidence from a cross-national twin study. *Journal of Personality and Social Psychology*, *121*(5), 1079–1094. https://doi.org/10 .1037/pspp0000366
- Kohler, H. P., Rodgers, J. L., & Christensen, K. (2002). Between nurture and nature: The shifting determinants of female fertility in Danish twin cohorts. *Social Biology*, 49(3–4), 218–248. https:// doi.org/10.1080/19485565.2002.9989060
- Kolk, M., & Barclay, K. (2021). Do income and marriage mediate the relationship between cognitive ability and fertility? Data from Swedish taxation and conscriptions registers for men born 1951– 1967. *Intelligence*, 84, Article 101514. https:// doi.org/10.1016/j.intell.2020.101514
- Krebs, J. R., & Davies, N. B. (1978). Behavioural ecology: An evolutionary approach. Blackwell Science.
- Krueger, R. F., South, S., Johnson, W., & Iacono, W. (2008). The heritability of personality is not always 50%: Gene-environment interactions and correlations between personality and parenting. *Journal* of *Personality*, 76(6), 1485–1522. https://doi.org/ 10.1111/j.1467-6494.2008.00529.x
- Li, N. P., van Vugt, M., & Colarelli, S. M. (2018). The evolutionary mismatch hypothesis: Implications for psychological science. *Current Directions in Psychological Science*, 27(1), 38–44. https://doi.org/ 10.1177/0963721417731378
- Lilienfeld, S. O., & Andrews, B. P. (1996). Development and preliminary validation of a self-report measure of psychopathic personality traits in noncriminal populations. *Journal of Personality Assessment*, 66(3), 488–524. https://doi.org/10 .1207/s15327752jpa6603\_3
- Manolio, T. A., Collins, F. S., Cox, N. J., Goldstein, D. B., Hindorff, L. A., Hunter, D. J., McCarthy, M. I., Ramos, E. M., Cardon, L. R., Chakravarti, A., Cho, J. H., Guttmacher, A. E., Kong, A., Kruglyak, L., Mardis, E., Rotimi, C. N., Slatkin, M., Valle, D., Whittemore, A. S., ... Visscher, P. M. (2009). Finding the missing heritability of complex diseases. *Nature*, 461(7265), 747–753. https://doi.org/10 .1038/nature08494
- Manuck, S. B., & McCaffery, J. M. (2014). Gene–environment interaction. Annual Review of Psychology, 65(1), 41–70. https://doi.org/10.1146/ annurev-psych-010213-115100
- Mealey, L. (1995). The sociobiology of sociopathy: An integrated evolutionary model. *Behavioral and*

Brain Sciences, 18(3), 523–541. https://doi.org/10 .1017/S0140525X00039595

- Mededović, J. (2017). Intelligence and fitness: The mediating role of educational level. *Evolutionary Psychology*, 15(2), Article 1474704917706936. https://doi.org/10.1177/1474704917706936
- Međedović, J. (2019a). Life history in a postconflict society: Violent intergroup conflict facilitates fast life-history strategy. *Human Nature*, 30(1), 59–70. https://doi.org/10.1007/s12110-018-09336-y
- Međedović, J. (2019b). Complex relations between psychopathy and fitness may indicate adaptive trade-offs. *Evolutionary Psychological Science*, 5(3), 257–266. https://doi.org/10.1007/s40806-018-00183-9
- Mededović, J. (2019c). Harsh environment facilitates psychopathy's Involvement in mating-parenting trade-off. *Personality and Individual Differences*, *139*, 235–240. https://doi.org/10.1016/j.paid.2018 .11.034
- Međedović, J. (2020). On the incongruence between psychometric and psychosocial-biodemographic measures of life history. *Human Nature*, *31*(3), 341–360. https://doi.org/10.1007/s12110-020-09377-2
- Međedović, J. (2021). Human life histories as dynamic networks: Using network analysis to conceptualize and analyze life history data. *Evolutionary Psychological Science*, 7(1), 76–90. https://doi.org/10 .1007/s40806-020-00252-y
- Mededović, J., & Petrović, B. (2019). Quantity-quality trade-offs may partially explain inter-individual variation in psychopathy. *Adaptive Human Behavior* and Physiology, 5(2), 211–226. https://doi.org/10 .1007/s40750-019-00113-4
- Međedović, J., Petrović, B., Želeskov-Đorić, J., & Savić, M. (2017). Interpersonal and affective psychopathy traits can enhance human fitness. *Evolutionary Psychological Science*, 3(4), 306–315. https://doi.org/10.1007/s40806-017-0097-5
- Međedović, J., Šoljaga, M., Stojković, A., & Gojević, I. (2018). Revealing complex relations between personality and fitness: HEXACO personality traits, life-time reproductive success and the age at first birth. *Personality and Individual Differences*, 129, 143–148. https://doi.org/10.1016/j.paid.2018.03 .014
- Meek, S. E., Lemery-Chalfant, K., Jahromi, L. B., & Valiente, C. (2013). A review of gene–environment correlations and their implications for autism: A conceptual model. *Psychological Review*, *120*(3), 497–521. https://doi.org/10.1037/a0033139
- Meij, J. J., Van Bodegom, D., Ziem, J. B., Amankwa, J., Polderman, A. M., Kirkwood, T. B. L., de Craen, A. J. M., Zwaan, B. J., & Westendorp, R. G. J. (2009). Quality–quantity trade-off of human offspring under adverse environmental conditions. *Journal of Evolutionary Biology*, 22(5), 1014–

1023. https://doi.org/10.1111/j.1420-9101.2009 .01713.x

- Mell, H., Safra, L., Algan, Y., Baumard, N., & Chevallier, C. (2018). Childhood environmental harshness predicts coordinated health and reproductive strategies: A cross-sectional study of a nationally representative sample from France. *Evolution and Human Behavior*, 39(1), 1–8. https://doi.org/10 .1016/j.evolhumbehav.2017.08.006
- Miller, W. B., Bard, D. E., Pasta, D. J., & Rodgers, J. L. (2010). Biodemographic modeling of the links between fertility motivation and fertility outcomes in the NLSY79. *Demography*, 47(2), 393– 414. https://doi.org/10.1353/dem.0.0107
- Mills, M. C., & Tropf, F. C. (2015). The biodemography of fertility: A review and future research frontiers. Kölner Zeitschrift für Soziologie und Sozialpsychologie, 67(Suppl 1), 397–424. https:// doi.org/10.1007/s11577-015-0319-4
- Milne, B. J., Moffitt, T. E., Crump, R., Poulton, R., Rutter, M., Sears, M. R., Taylor, A., & Caspi, A. (2008). How should we construct psychiatric family history scores? A comparison of alternative approaches from the Dunedin Family Health History Study. *Psychological Medicine*, 38(12), 1793–1802. https:// doi.org/10.1017/S0033291708003115
- Milot, E., Mayer, F. M., Nussey, D. H., Boisvert, M., Pelletier, F., & Réale, D. (2011). Evidence for evolution in response to natural selection in a contemporary human population. *Proceedings of the National Academy of Sciences of the United States* of America, 108(41), 17040–17045. https:// doi.org/10.1073/pnas.1104210108
- Nettle, D., & Bateson, M. (2015). Adaptive developmental plasticity: What is it, how can we recognize it and when can it evolve? *Proceedings of the Royal Society B: Biological Sciences*, 282(1812), Article 20151005. https://doi.org/10.1098/rspb.2015.1005
- Pasta, D. J., & Miller, W. B. (2000). A heritability study of childbearing motivation. In J. L. Rodgers, D. C. Rowe, & W. B. Miller (Eds.), *Genetic influences on human fertility and sexuality* (pp. 107– 120). Springer.
- Patrick, C. J., Fowles, D. C., & Krueger, R. F. (2009). Triarchic conceptualization of psychopathy: Developmental origins of disinhibition, boldness, and meanness. *Development and Psychopathology*, 21(3), 913–938. https://doi.org/10.1017/ S0954579409000492
- Penke, L., & Jokela, M. (2016). The evolutionary genetics of personality revisited. *Current Opinion in Psychology*, 7, 104–109. https://doi.org/10.1016/j .copsyc.2015.08.021
- Pettay, J. E., Charmantier, A., Wilson, A. J., & Lummaa, V. (2008). Age-specific genetic and maternal effects in fecundity of preindustrial Finnish women. *Evolution: International Journal of Organic*

*Evolution*, 62(9), 2297–2304. https://doi.org/10 .1111/j.1558-5646.2008.00452.x

- Pettay, J. E., Kruuk, L. E., Jokela, J., & Lummaa, V. (2005). Heritability and genetic constraints of lifehistory trait evolution in preindustrial humans. *Proceedings of the National Academy of Sciences of the United States of America*, 102(8), 2838–2843. https://doi.org/10.1073/pnas.0406709102
- Plomin, R., DeFries, J. C., & Loehlin, J. C. (1977). Genotype–environment interaction and correlation in the analysis of human behavior. *Psychological Bulletin*, 84(2), 309–322. https://doi.org/10.1037/ 0033-2909.84.2.309
- Plomin, R., DeFries, J. C., McClearn, G. E., & McGuffin, P. (2008). *Behavioral genetics* (5th ed.). Worth Publishers.
- Polderman, T. J., Benyamin, B., De Leeuw, C. A., Sullivan, P. F., Van Bochoven, A., Visscher, P. M., & Posthuma, D. (2015). Meta-analysis of the heritability of human traits based on fifty years of twin studies. *Nature Genetics*, 47(7), 702–709. https:// doi.org/10.1038/ng.3285
- Posthuma, D., Beem, A. L., De Geus, E. J., Van Baal, G. C. M., Von Hjelmborg, J. B., Iachine, I., & Boomsma, D. I. (2003). Theory and practice in quantitative genetics. *Twin Research and Human Genetics*, 6(5), 361–376. https://doi.org/10.1375/ 136905203770326367
- Purcell, S. (2002). Variance components models for gene–environment interaction in twin analysis. *Twin Research and Human Genetics*, 5(6), 554– 571. https://doi.org/10.1375/136905202762342026
- Reeve, C. L., Lyerly, J. E., & Peach, H. (2013). Adolescent intelligence and socio-economic wealth independently predict adult marital and reproductive behavior. *Intelligence*, 41(5), 358–365. https:// doi.org/10.1016/j.intell.2013.05.010
- Rodgers, J. L., Bard, D. E., & Miller, W. B. (2007). Multivariate Cholesky models of human female fertility patterns in the NLSY. *Behavior Genetics*, 37(2), 345–361. https://doi.org/10.1007/s10519-006-9137-9
- Rodgers, J. L., Kohler, H. P., Kyvik, K. O., & Christensen, K. (2001). Behavior genetic modeling of human fertility: Findings from a contemporary Danish twin study. *Demography*, 38(1), 29–42. https:// doi.org/10.1353/dem.2001.0009
- Roff, D. A. (2002). *Life history evolution*. Sinauer Associates.
- Rubin, H. (2016). The phenotypic gambit: Selective pressures and ESS methodology in evolutionary game theory. *Biology & Philosophy*, 31(4), 551– 569. https://doi.org/10.1007/s10539-016-9524-4
- Ruby, J. G., Wright, K. M., Rand, K. A., Kermany, A., Noto, K., Curtis, D., Varner, N., Garrigan, D., Slinkov, D., Dorfman, I., Granka, J. M., Byrnes, J., Myres, N., & Ball, C. (2018). Estimates of the heritability of human longevity are substantially inflated

due to assortative mating. *Genetics*, 210(3), 1109–1124. https://doi.org/10.1534/genetics.118.301613

- Sanjak, J. S., Sidorenko, J., Robinson, M. R., Thornton, K. R., & Visscher, P. M. (2018). Evidence of directional and stabilizing selection in contemporary humans. *Proceedings of the National Academy of Sciences of the United States of America*, 115(1), 151–156. https://doi.org/10.1073/pnas.1707227114
- Sauce, B., & Matzel, L. D. (2018). The paradox of intelligence: Heritability and malleability coexist in hidden gene–environment interplay. *Psychological Bulletin*, 144(1), 26–47. https://doi.org/10.1037/ bul0000131
- Schwartz, J. A., Connolly, E. J., Beaver, K. M., Nedelec, J. L., & Vaughn, M. G. (2015). Proposing a pedigree risk measurement strategy: Capturing the intergenerational transmission of antisocial behavior in a nationally representative sample of adults. *Twin Research and Human Genetics*, 18(6), 772– 784. https://doi.org/10.1017/thg.2015.68
- Sear, R. (2020). Do human "life history strategies" exist? *Evolution and Human Behavior*, *41*(6), 513– 526. https://doi.org/10.1016/j.evolhumbehav.2020 .09.004
- Shanahan, M. J., & Hofer, S. M. (2005). Social context in gene–environment interactions: Retrospect and prospect. *The Journals of Gerontology: Series B*, 60(Special\_Issue\_1), 65–76. https://doi.org/10 .1093/geronb/60.Special\_Issue\_1.65
- Shatz, S. M. (2008). IQ And fertility: A cross-national study. *Intelligence*, 36(2), 109–111. https://doi.org/ 10.1016/j.intell.2007.03.002
- Sheppard, P., Pearce, M. S., & Sear, R. (2016). How does childhood socioeconomic hardship affect reproductive strategy? Pathways of development. *American Journal of Human Biology*, 28(3), 356– 363. https://doi.org/10.1002/ajhb.22793
- Silventoinen, K., Helle, S., Nisen, J., Martikainen, P., & Kaprio, J. (2013). Height, age at first birth, and lifetime reproductive success: A prospective cohort study of Finnish male and female twins. *Twin Research and Human Genetics*, *16*(2), 581–589. https://doi.org/10.1017/thg.2012.150
- Sinervo, B. & Calsbeek, R. (2010). Behavioral concepts of selection. In D. F. Westneat & C. W. Fox (Eds). *Evolutionary behavioural ecology* (pp. 32–45). Oxford University Press.
- Smith, B. R., & Blumstein, D. T. (2008). Fitness consequences of personality: A meta-analysis. *Behavio*ral Ecology, 19(2), 448–455. https://doi.org/10 .1093/beheco/arm144
- Sodini, S. M., Kemper, K. E., Wray, N. R., & Trzaskowski, M. (2018). Comparison of genotypic and phenotypic correlations: Cheverud's Conjecture in humans. *Genetics*, 209(3), 941–948. https:// doi.org/10.1534/genetics.117.300630
- Sørensen, K., Juul, A., Christensen, K., Skytthe, A., Scheike, T., & Kold Jensen, T. (2013). Birth size

and age at menarche: A twin perspective. *Human Reproduction*, 28(10), 2865–2871. https://doi.org/ 10.1093/humrep/det283

- Stearns, S. C., Byars, S. G., Govindaraju, D. R., & Ewbank, D. (2010). Measuring selection in contemporary human populations. *Nature Reviews Genetics*, 11(9), 611–622. https://doi.org/10.1038/ nrg2831
- Stearns, S. C., & Rodrigues, A. M. (2020). On the use of "life history theory" in evolutionary psychology. *Evolution and Human Behavior*, 41(6), 474–485. https://doi.org/10.1016/j.evolhumbehav.2020.02 .001
- Sulem, P., Gudbjartsson, D. F., Rafnar, T., Holm, H., Olafsdottir, E. J., Olafsdottir, G. H., Jonsson, T., Alexandersen, P., Feenstra, B., Boyd, H. A., Aben, K. K., Verbeek, A. L., Roeleveld, N., Jonasdottir, A., Styrkarsdottir, U., Steinthorsdottir, V., Karason, A., Stacey, S. N., Gudmundsson, J., ... Stefansson, K. (2009). Genome-wide association study identifies sequence variants on 6q21 associated with age at menarche. *Nature Genetics*, 41(6), 734–738. https://doi.org/10.1038/ng.383
- Tropf, F. C., Barban, N., Mills, M. C., Snieder, H., & Mandemakers, J. J. (2015). Genetic influence on age at first birth of female twins born in the UK, 1919–68. *Population Studies*, 69(2), 129–145. https://doi.org/10.1080/00324728.2015.1056823
- Tropf, F. C., Stulp, G., Barban, N., Visscher, P. M., Yang, J., Snieder, H., & Mills, M. C. (2015). Human fertility, molecular genetics, and natural selection in modern societies. *PLoS ONE*, *10*(6), Article e0126821. https://doi.org/10.1371/journal .pone.0126821
- Trumbetta, S. L., Markowitz, E. M., & Gottesman, I. I. (2007). Marriage and genetic variation across the lifespan: Not a steady relationship? *Behavior Genetics*, 37(2), 362–375. https://doi.org/10.1007/ s10519-006-9132-1
- Tucker-Drob, E. M., & Bates, T. C. (2016). Large cross-national differences in gene × socioeconomic status interaction on intelligence. *Psychological Science*, 27(2), 138–149. https://doi.org/10.1177/ 0956797615612727
- Tucker-Drob, E. M., Briley, D. A., & Harden, K. P. (2013). Genetic and environmental influences on cognition across development and context. *Current Directions in Psychological Science*, 22(5), 349– 355. https://doi.org/10.1177/0963721413485087
- Turkheimer, E., Haley, A., Waldron, M., D'Onofrio, B., & Gottesman, I. I. (2003). Socioeconomic status modifies heritability of IQ in young children. *Psychological Science*, 14(6), 623–628. https:// doi.org/10.1046/j.0956-7976.2003.psci\_1475.x
- Tuvblad, C., Wang, P., Patrick, C. J., Berntsen, L., Raine, A., & Baker, L. A. (2019). Genetic and environmental influences on disinhibition, boldness, and meanness as assessed by the triarchic psychopathy

measure in 19–20-year-old twins. *Psychological Medicine*, 49(9), 1500–1509. https://doi.org/10.1017/S0033291718002052

- Udry, J. R. (1996). Biosocial models of low-fertility societies. *Population and Development Review*, 22(Suppl.), 325–336. https://doi.org/10.2307/ 2808017.
- van den Berg, S. M., & Boomsma, D. I. (2007). The familial clustering of age at menarche in extended twin families. *Behavior Genetics*, 37(5), 661–667. https://doi.org/10.1007/s10519-007-9161-4
- van Oers, K., & Sinn, D. L. (2011). Toward a basis for the phenotypic gambit: Advances in the evolutionary genetics of animal personality. In M. Inoue-Murayama, S. Kawamura, & A. Weiss (Eds.), From genes to animal behavior, primatology monographs (pp. 165–183). Springer.
- Visscher, P. M., Hill, W. G., & Wray, N. R. (2008). Heritability in the genomics era—Concepts and misconceptions. *Nature Reviews Genetics*, 9(4), 255–266. https://doi.org/10.1038/nrg2322
- Vukasović, T., & Bratko, D. (2015). Heritability of personality: A meta-analysis of behavior genetic studies. *Psychological Bulletin*, 141(4), 769–785. https://doi.org/10.1037/bul0000017
- Waldron, M., Heath, A. C., Turkheimer, E., Emery, R., Bucholz, K. K., Madden, P. A., & Martin, N. G. (2007). Age at first sexual intercourse and teenage pregnancy in Australian female twins. *Twin Research and Human Genetics*, 10(3), 440–449. https://doi.org/10.1375/twin.10.3.440
- Waldron, M., Heath, A. C., Turkheimer, E. N., Emery, R. E., Nelson, E., Bucholz, K. K., Madden, P. A., & Martin, N. G. (2008). Childhood sexual abuse moderates genetic influences on age at first consensual sexual intercourse in women. *Behavior Genetics*, 38(1), 1–10. https://doi.org/10.1007/s10519-007-9176-x
- Walsh, B., & Blows, M. W. (2009). Abundant genetic variation + strong selection = multivariate genetic constraints: A geometric view of adaptation. Annual Review of Ecology, Evolution, and Systematics,

40(1), 41–59. https://doi.org/10.1146/annurev .ecolsys.110308.120232

- Webster, G. D., Graber, J. A., Gesselman, A. N., Crosier, B. S., & Schember, T. O. (2014). A life history theory of father absence and menarche: A metaanalysis. *Evolutionary Psychology*, 12(2), 273– 294. https://doi.org/10.1177/147470491401200202
- Wilson, M., & Daly, M. (1997). Life expectancy, economic inequality, homicide, and reproductive timing in Chicago neighbourhoods. *BMJ*, 314(7089), 1271–1274. https://doi.org/10.1136/bmj.314.7089 .1271
- Wolf, J. B., & Moore, A. J. (2010). Interacting phenotypes and indirect genetic effects: A genetic perspective on the evolution of social behavior. In D. F. Westneat & C. W. Fox (Eds.), *Evolutionary behavioral ecology* (pp. 225–245). Oxford University Press.
- Xu, Y., Norton, S., & Rahman, Q. (2018). Early life conditions, reproductive and sexuality-related life history outcomes among human males: A systematic review and meta-analysis. *Evolution and Human Behavior*, 39(1), 40–51. https://doi.org/10 .1016/j.evolhumbehav.2017.08.005
- Yang, J., Lee, S. H., Goddard, M. E., & Visscher, P. M. (2011). GCTA: A tool for genome-wide complex trait analysis. *The American Journal of Human Genetics*, 88(1), 76–82. https://doi.org/10.1016/j .ajhg.2010.11.011
- Zietsch, B. P., de Candia, T. R., & Keller, M. C. (2015). Evolutionary behavioral genetics. *Current Opinion in Behavioral Sciences*, 2, 73–80. https:// doi.org/10.1016/j.cobeha.2014.09.005
- Zietsch, B. P., & Sidari, M. J. (2020). A critique of life history approaches to human trait covariation. *Evolution and Human Behavior*, 41(6), 527–535. https://doi.org/10.1016/j.evolhumbehav.2019.05 .007

Received July 1, 2022 Revision received November 17, 2022 Accepted November 23, 2022